Author's response to reviews

Title: Identification of losses to follow-up in a community-based antiretroviral therapy clinic in South Africa using a computerized pharmacy tracking system

Authors:

Mweete D Nglazi (Mweete.Nglazi@hiv-research.org.za)
Richard Kaplan (richard.kaplan@hiv-research.org.za)
Robin Wood (robin.wood@hiv-research.org.za)
Linda-Gail Bekker (linda-gail.bekker@hiv-research.org.za)
Stephen D Lawn (stevelawn@yahoo.co.uk)

Version: 8 Date: 5 November 2010

Author's response to reviews: see over
Dear Editor,

Thank you for your email of the 29th October 2010. We have proofread the manuscript and made the following minor changes to the manuscript:

**Minor revisions made:**

Authors contributions: The authors initials have been changed accordingly in keeping with the changes made by your team on page 1.

Acknowledgements: The authors initials have been changed accordingly in keeping with the changes made by your team on page 1.

We thank the Biomed Central Editorial Team for helping us to make the manuscript conform to the journal style. We confirm that the revised version of this manuscript is acceptable for publication. We look forward to your feedback in due course.

Yours sincerely,

Mweete D. Nglazi, on behalf of the co-authors